Author's response to reviews

Title: Pharmacological inhibition of poly(ADP-ribose) polymerase-1 modulates resistance of human glioblastoma stem cells to temozolomide

Authors:

Lucio Tentori (tentori@uniroma2.it)
Lucia Ricci-Vitiani (lriccivitiani@yahoo.it)
Alessia Muzi (alessia.muzi@libero.it)
Fabio Ciccarone (Ciccarone@bce.uniroma1.it)
Federica Pelacchi (pelacchifederica@yahoo.it)
Roberta Calabrese (r.calabrese@bce.uniroma1.it)
Daniele Runci (daniele.talenti@hotmail.it)
Roberto Pallini (pallini@rm.unicatt.it)
Paola Caiafa (caiafa@bce.uniroma1.it)
Grazia Graziani (graziani@uniroma2.it)

Version: 3
Date: 19 February 2014

Author's response to reviews:

Dear Editor:

as requested the manuscript has been modified adding:

- the e-mails of all authors, in the title page;

- a sentence stating that all patients provided written informed consent according to the Declaration of Helsinki and that the research proposal was approved by the Ethical Committee of the Università Cattolica del Sacro Cuore (Rome, Italy), in the Materials and Methods section (page 7).